Author's response to reviews

Title: Using the Ages and Stages Questionnaire to teach medical students developmental assessment: a descriptive analysis

Authors:

Pam A Nicol (pam.nicol@uwa.edu.au)

Version: 5 Date: 10 March 2006

Author's response to reviews:

Thank you for your comments. All the punctuation, suggested amendments and referencing revisions have been completed. The test for the 'student confidence' was in methods (Wilcoxon Mann-Whitney Test, p>0.05) but has now also been added to results. The comparison between the ASQ and student report findings were not analysed statistically. It was reported as percentage agreement, as used in the original ASQ interobserver reliability analysis between parents and expert examiners (94%, standard error 0.12).